

## GOPEN ACCESS

**Citation:** Docken SS, McCormick K, Pampena MB, Samer S, Lindemuth E, Pinkevych M, et al. (2023) Preferential selection of viral escape mutants by CD8<sup>+</sup> T cell 'sieving' of SIV reactivation from latency. PLoS Pathog 19(11): e1011755. https:// doi.org/10.1371/journal.ppat.1011755

**Editor:** Robert F. Siliciano, Johns Hopkins University School of Medicine, UNITED STATES

Received: June 30, 2023

Accepted: October 15, 2023

Published: November 30, 2023

**Peer Review History:** PLOS recognizes the benefits of transparency in the peer review process; therefore, we enable the publication of all of the content of peer review and author responses alongside final, published articles. The editorial history of this article is available here: https://doi.org/10.1371/journal.ppat.1011755

**Copyright:** This is an open access article, free of all copyright, and may be freely reproduced, distributed, transmitted, modified, built upon, or otherwise used by anyone for any lawful purpose. The work is made available under the <u>Creative</u> Commons CC0 public domain dedication.

Data Availability Statement: Data and code used in the current study are available at the GitHub repository <u>https://github.com/iap-sydney/CD8\_</u> sieving\_TatSL8\_PlosPath2023. **RESEARCH ARTICLE** 

# Preferential selection of viral escape mutants by CD8<sup>+</sup> T cell 'sieving' of SIV reactivation from latency

Steffen S. Docken<sup>1</sup>, Kevin McCormick<sup>2</sup>, M. Betina Pampena<sup>2,3</sup>, Sadia Samer<sup>4</sup>, Emily Lindemuth<sup>5</sup>, Mykola Pinkevych<sup>1</sup>, Elise G. Viox<sup>4</sup>, Yuhuang Wu<sup>1</sup>, Timothy E. Schlub<sup>6</sup>, Deborah Cromer<sup>1</sup>, Brandon F. Keele<sup>7</sup>, Mirko Paiardini<sup>4,8</sup>, Michael R. Betts<sup>2,3</sup>, Katharine J. Bar<sup>5</sup>, Miles P. Davenport<sup>1</sup>\*

 Infection Analytics Program, Kirby Institute, UNSW Sydney, Sydney, New South Wales, Australia,
 Department of Microbiology, Perelman School of Medicine, University of Pennsylvania, Philadelphia, Pennsylvania, United States of America, 3 Center for AIDS Research and Institute for Immunology, Perelman School of Medicine, University of Pennsylvania, Philadelphia, Pennsylvania, United States of America,
 Division of Microbiology and Immunology, Emory National Primate Research Center, Emory University, Atlanta, Georgia, United States of America, 5 Department of Medicine, University of Pennsylvania, Philadelphia, Pennsylvania, United States of America, 6 Sydney School of Public Health, Faculty of Medicine and Health, University of Sydney, Sydney, New South Wales, Australia, 7 AIDS and Cancer Virus Program, Frederick National Laboratory for Cancer Research, Frederick, Maryland, United States of America,
 Department of Pathology and Laboratory Medicine, School of Medicine, Emory University, Atlanta, Georgia, United States of America

\* m.davenport@unsw.edu.au

## Abstract

HIV rapidly rebounds after interruption of antiretroviral therapy (ART). HIV-specific CD8<sup>+</sup> T cells may act to prevent early events in viral reactivation. However, the presence of viral immune escape mutations may limit the effect of CD8<sup>+</sup> T cells on viral rebound. Here, we studied the impact of CD8 immune pressure on post-treatment rebound of barcoded SIVmac293M in 14 Mamu-A\*01 positive rhesus macagues that initiated ART on day 14, and subsequently underwent two analytic treatment interruptions (ATIs). Rebound following the first ATI (seven months after ART initiation) was dominated by virus that retained the wildtype sequence at the Mamu-A\*01 restricted Tat-SL8 epitope. By the end of the two-month treatment interruption, the replicating virus was predominantly escaped at the Tat-SL8 epitope. Animals reinitiated ART for 3 months prior to a second treatment interruption. Time-torebound and viral reactivation rate were significantly slower during the second treatment interruption compared to the first. Tat-SL8 escape mutants dominated early rebound during the second treatment interruption, despite the dominance of wild-type virus in the proviral reservoir. Furthermore, the escape mutations detected early in the second treatment interruption were well predicted by those replicating at the end of the first, indicating that escape mutant virus in the second interruption originated from the latent reservoir as opposed to evolving de novo post rebound. SL8-specific CD8<sup>+</sup> T cell levels in blood prior to the second interruption were marginally, but significantly, higher (median 0.73% vs 0.60%, p = 0.016). CD8<sup>+</sup> T cell depletion approximately 95 days after the second treatment interruption led to the reappearance of wild-type virus. This work suggests that CD8<sup>+</sup> T cells can actively

Funding: This work was supported in part by NIH grants NIAID P01AI131338 (to MPD, KJB, MRB, and MP); NHLBI, NIDDK, NIMH, NINDS, NIDA, and NIAID 1UM1AI164562-01 (to MP, MPD, and BFK); NIMH R01 MH128155, NIAID P30-AI-045008, R01 AI162646, and UM1AI164570 (to KJB). MPD is supported by an NHMRC Investigator grant (1173027) and an NHMRC Program grant (149990). This project has been funded in part with federal funds from the National Cancer Institute, National Institutes of Health, under Contract No. 75N91019D00024/HHSN261201500003I (BFK). The content of this publication does not necessarily reflect the views or policies of the Department of Health and Human Services, nor does mention of trade names, commercial products, or organizations imply endorsement by the U.S. Government. The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

**Competing interests:** Michael R. Betts is a paid consultant of Interius Biotherapeutics. The authors have no other competing interests to declare.

suppress the rebound of wild-type virus, leading to the dominance of escape mutant virus after treatment interruption.

## Author summary

During untreated HIV infection, CD8<sup>+</sup> ('killer') T cells target and apply selective pressure on the HIV virus. Unfortunately, immune escape mutation may occur, allowing HIV to evade immune recognition. HIV may be more susceptible to immune pressure when there are low levels of virus, such as during initial viral rebound following treatment interruption. However, it is unknown if the immune system can target and suppress the growth of rebounding virus. In this study, we demonstrate that CD8<sup>+</sup> T cells can inhibit the reactivation of susceptible virus following treatment interruption. However, escaped virus can circumvent CD8 pressure and reactivate if it is present in the latent viral reservoir.

## Introduction

HIV viral replication can be suppressed by ART. However, interruption of treatment leads to rebound of replicating virus in most individuals. This rebound arises from reactivation of individual 'reactivation founder' viral strains, and the frequency of reactivation from latency has been estimated to range from around once a week in humans, to several times a day in SIV-infected macaques [1–3]. This reactivation of a small number of viral clonotypes creates a population bottleneck that is potentially susceptible to immune control. Treatment with monoclonal broadly neutralizing antibodies (bNAbs) has been shown to delay viral reactivation in the absence of ART [4,5]. However, the presence of viral escape mutation limits the ability of passive antibody therapy to suppress viral replication following ATI [5–7]. Antibody escape mutants may arise from variants present in the latent proviral reservoir, and bNAbs present at the time of treatment interruption act by 'sieving' the reactivation of bNAb-sensitive virus, which acquires escape mutations during post-rebound replication.

It is well established that CD8<sup>+</sup> T cells drive immune escape during both the acute and chronic phases of HIV and SIV infection [8-13]. In the viremic SIV model, immune escape can occur very rapidly at the Mamu A\*01 restricted Tat-S(T)L8 T cell epitope during the acute phase of infection [14-16]. While immune escape during fully suppressive therapy has not been demonstrated in the SIV model, studies in ART-treated people living with HIV (PLWH) have suggested that CD8<sup>+</sup> T cell-driven viral evolution may occur [17,18]. After ATI in humans, it is also established that expansion of memory HIV-specific CD8<sup>+</sup> T cell responses closely mirrors viral recrudescence [19-21]. Despite this association, little is known about the precise role and impact of T cell pressure on viral reactivation and growth after ATI. Previously, Okoye et al. showed that SIV-infected rhesus macaques (RMs) that underwent CD8<sup>+</sup> T cell depletion at ATI had higher peak and set point viral loads during viral rebound than untreated controls [22], but no difference in time-to-rebound or initial growth rate of virus following treatment interruption was observed in the CD8-depleted animals. This suggests that CD8 pressure in a native, unaugmented state does not develop until after reactivation has been initiated and is unable to control early reactivation events. However, it is also important to note that the RMs in this study initiated ART early at 12 days post infection (dpi), and thus may not have developed sufficiently potent CD8<sup>+</sup> T cell responses to suppress early events in

viral reactivation [23]. Thus, whether CD8<sup>+</sup> T cell responses are able to suppress the reactivation or early replication of wild-type (WT) virus and act as a 'sieve' for selection of viral escape mutation has not been determined.

Here, we study the impact of CD8<sup>+</sup> T cell pressure on immune escape following treatment interruption in a non-human primate model using barcoded SIVmac239M [3] that allows tracking of the dynamics of individual viral lineages. The barcode site is just upstream of the Tat-SL8 epitope, which has been well characterized as a locus of CD8 pressure early in SIV infection of Mamu<sup>\*</sup>A01<sup>+</sup> RMs [14-16]. Tat-SL8-specific CD8<sup>+</sup> T cells have routinely been identified through single peptide stimulation, indicating the strong specificity of these cells [14-16], and it has been shown that addition of Tat-SL8 specific CD8<sup>+</sup> T cells substantially inhibits SIVmac239 replication in vitro [16]. Due to the proximity of Tat-SL8 to the barcoded region of SIVmac239M, both can be sequenced simultaneously via high throughput Illumina sequencing, allowing studies of immune escape on individual viral barcode lineages. Using this strategy, we assessed the evolution of immune escape at the SL8 epitope in Mamu $^{*}A01^{+}$ RMs over the course of two periods of antiretroviral treatment followed by two treatment interruptions. We found low levels of immune escape prior to ART at 14 dpi, as well as in cellassociated SIV DNA sampled during the first period of ART suppression. After the first ATI, early rebound virus was composed of predominantly WT virus. However, we observed extensive mutation within the SL8 epitope during the first ATI. A very different pattern of viral mutation was observed during the second ATI, where escape mutant virus was dominant from the earliest timepoints. Analysis of viral barcodes and SL8 epitope mutations, along with characterisation of CD8<sup>+</sup> T cell responses, indicated that CD8<sup>+</sup> T cells limited the reactivation or early replication events of WT virus after the second treatment interruption. To our knowledge, this is the first evidence that CD8<sup>+</sup> T cells are capable of inhibiting early events in the reactivation of susceptible latent virus.

#### Results

#### Animal experiment

14 adult male Mamu A\*01<sup>+</sup> RMs (average age 6 years) were intravenously infected with  $10^4$  TCID50 SIVmac239M [3] and achieved peak viral loads of between 6.9 and 8.06 log<sub>10</sub> *copies/ ml*. At 14 dpi, RMs initiated daily ART (Fig 1A). Viral load was suppressed below the detection threshold of 60 copies/ml by 81 dpi in all but two animals and by 113 dpi in all animals.

ART was withdrawn on day ~220, and 13 animals had detectable rebounding virus between 10 and 25 days after ART withdrawal, while animal RNd15 did not have a detectable rebound until a single detection of virus at per-protocol ART re-initiation (61 days post ATI). Animals rebounded to mean log<sub>10</sub> peak viral load of 4.54 log<sub>10</sub> copies/ml (range 3.54–6.02), and then achieved a mean viral set point of 3.42 log<sub>10</sub> copies/ml between days 30 and up to 60 post rebound (range 1.96–4.45; see Analysis definitions in Materials and Methods for exact time frame). The animals analysed in this study were part of a larger cohort used to determine the immunological effects of treatment with FTY720, a drug that arrests T cell migration from tissues [24, 25]. Thus, half of the animals received FTY720 for 30 days prior and throughout the first ATI (see Fig 1A). Notably, FTY720 had no significant effect on any of the viral rebound parameters used for our analyses of viral escape dynamics (similar to as reported for FTY720 treatment during early ART [26]); as a result, the FTY720-treated and control animals were combined for this study.

Next, we estimated the frequency of successful reactivation from latency after ART interruption (Fig 1G). The median reactivation rate was 3.12 reactivation events per day across all 13 reactivating animals (range 0.35 to 10.64). For these calculations, viral growth rate and



**Fig 1. Experimental Protocol.** (A) Schematic representation of experimental protocol. 14 animals were infected with  $10^4$  TCID50 of SIVmac239M and began ART on day 14. ART treatment indicated by grey shading. 7 animals were treated during the first interruption with FTY720 and 7 animals were controls (indicated by dark green and light green bars, respectively). ART treatment was interrupted twice and 10 animals were CD8 depleted after the second treatment interruption. (B) As previously described [3], SIVmac239M was generated from the infectious molecular clone SIVmac239 by inserting barcodes between the vpx and vpr accessory genes, resulting in the barcodes lying 247 base pairs from the Tat-SL8 CD8 escape epitope. (C) Viral loads of FTY720-treated and control animals indicated by dark green and light green symbols, respectively. (D) Proportion of the circulating virus (dark green and light green circles for FTY720-treated and control) and LNMC and PBMC cell-associated DNA (squares and triangles, respectively; cyan and light cyan for FTY720-treated and control) that remains WT at the SL8 epitope measured by high throughput sequencing throughout the protocol. (E) Time-to-detection of viral rebound in first and second treatment interruptions (dark red and light red lines, respectively). P-value of logrank test provided. (F) Time course of percent of total CD8<sup>+</sup> T cells in PBMC that are specific for the Tat-SL8 epitope (dark green and light green circles

for FTY720-treated and control, respectively). (G) Reactivation rates based on barcode proportions (see Estimation of the reactivation rate in Materials and Methods) following first and second treatment interruptions (dark red and light red circles, respectively).

https://doi.org/10.1371/journal.ppat.1011755.g001

relative size of different viral barcodes within an animal were used to estimate delays between individual successful reactivations, and thereby estimate frequency of successful reactivation. This methodology (as outlined in our previous publications [2, 3] and briefly in Materials and Methods) is based on the assumptions that timing of successful reactivations within an animal are exponentially distributed and all barcodes grow at the same rate following reactivation.

A second round of ART was resumed after ~62 days of treatment interruption (at ~282 dpi), and viral load was resuppressed below detection in all animals by 21 days after ART reinitiation. Animals were maintained on the second ART treatment for 3 months until a second ATI was performed at 374 or 375 dpi. Time-to-detection of viral rebound was significantly longer at the second interruption (Fig 1E; p = 0.019). Additionally, peak viral levels during the second interruption were significantly lower than in the first interruption (median of 3.36 (n = 12) vs. 4.50  $(n = 13) \log_{10}$  copies/ml, p = 0.003), as were time weighted  $\log_{10}$  viral set points (median 2.20 (n = 8) vs. 3.62 (n = 13)  $\log_{10}$  copies/ml from day 30 post rebound up to day 60 (see Analysis definitions in Materials and Methods for exact time frame), p = 0.016). The frequency of successful reactivation from latency estimated using barcode analysis was around 6 times slower during the second interruption (Fig 1G), with median reactivation rate of 0.55 per day (vs. 3.12 in first ATI, p = 0.0039). A direct comparison of the number of successfully rebounding barcodes reveals the same qualitative result (p = 0.0003; see S2 Fig). We note that our sequencing only identified barcoded clonotypes that successfully expanded to the level of detection by sequencing. Therefore, it is possible many more barcodes initially reactivated from latency following ATI but subsequently failed to grow to the level of detection.

### Induction and maintenance of Tat-SL8 specific CD8<sup>+</sup> T cells

We next assessed the longitudinal frequency and phenotype of Tat-SL8 specific CD8<sup>+</sup> T lymphocytes in peripheral blood mononuclear cells (PBMC) via flow cytometry. We observed an induction of Tat-SL8 tetramer binding CD8<sup>+</sup> T lymphocytes following primary infection, with a median of 0.076% (range: 0.007-2.07%; Fig 1F) of CD8<sup>+</sup> T cells in PBMC specific for Tat-SL8 at day 14 post-infection. This was increased to a median of 0.60% by day ~190, although this was not significant (p = 0.078). The frequency of Tat-SL8 specific CD8<sup>+</sup> T cells in PBMCs of control animals increased significantly during the first treatment interruption compared to ~190 dpi (p = 0.016), reaching peak levels of 1.35% (range: 0.52-2.97%). Note that we only consider control animals in our analysis of CD8 dynamics during the first ATI due to FTY720's effect on T cell migration in treated animals. Following initiation of the second round of ART (day 281-282), the frequency of Tat-SL8-specific T cells declined from their peak during the first interruption. When comparing the frequency of Tat-SL8-specific cells in PBMC prior to each interruption, we observed that the responses were higher prior to the second interruption (day 373-374 vs. day ~190; median 0.73% vs. 0.60% in PBMC; p = 0.016). Deeper exploration of the phenotypes of Tat-SL8-specific CD8<sup>+</sup> T Cells in PBMC (S1 Table) revealed that although the proportion of proliferating (Ki-67<sup>+</sup>) and cytotoxic (Perforin<sup>+</sup>/granzyme B<sup>+</sup>) cells during ART were low, there was a slight decrease prior to the second ATI compared to before the first (p = 0.008 and 0.027, respectively). Thus, it appears that CD8<sup>+</sup> T cell responses were very similar prior to the second interruption compared to the first. However, we note this is based on analysis of cells from peripheral blood samples, and it is unclear if there was a difference in the functional profile of CD8<sup>+</sup> T cells in tissue.

#### Immune escape during primary infection and treatment interruption

To investigate the development of immune escape from recognition by SL8-specific CD8<sup>+</sup> T cells we sequenced the Tat-SL8 epitope in plasma virus and cell-associated DNA, defining all non-WT amino acid sequences as escape variants. Prior to ART initiation (14 dpi), we saw limited evidence of viral escape mutations (range 4.8–18% plasma virus was non-wild type at the SL8 epitope, circles in Fig 1D). This is consistent with previous work [27] indicating low levels of escape variants over the first 14 days of infection. Proviral cell-associated DNA in LNMC and PBMC was also analysed during the first round of ART. This analysis revealed lower levels of escape mutant virus in both LNMC and PBMC cell-associated DNA than plasma virus at time of treatment (6.0% or less in all animals at day 185–220; n = 12 for both LNMC and PBMC (squares and triangles, respectively, in Fig 1D)). This is consistent with there being very little selection for escape prior to or during the initial ART treatment, and relatively low levels of escape in the cell-associated DNA on ART.

Following the first treatment interruption, early rebounding virus (<10 days after rebound detection; n = 10) was predominantly wild-type (median 12% escape, range 3.3–55%) and the degree of escape was not significantly different from that observed during primary infection (p = 1). However, the proportion of escape mutant virus increased throughout the first ATI (Fig 2A dark red symbols), and by the time ART was re-initiated (after ~60 days of treatment



**Fig 2. Immune escape dynamics. (A)** Percentage of overall viral load in each animal that is wild-type at the Tat-SL8 epitope plotted vs. days post infection (black) or days after reactivation (dark and light red for interruptions 1 and 2, respectively). Control and FTY720-treated animals are indicated by circles and triangles, respectively. **(B)** Percentage of overall viral load in each animal that is wild-type at the Tat-SL8 epitope plotted vs. total days of replication (primary infection + number of days since detection in the first interruption). Time points from primary infection and 1<sup>st</sup> ATI again plotted in black and dark red, respectively. Data from Immonen et al. 2020 [27] and Allen et al. 2000 [14] are also plotted in dark and light orange, respectively (background subtraction not applied to data from previous studies and open symbols indicate limit of detection at time points when wild type was not detected in these studies).

https://doi.org/10.1371/journal.ppat.1011755.g002

interruption, 36–53 days post reactivation) we observed that an average 85% of plasma virus was escaped within the Tat-SL8 region (range 38–99%; n = 11). We also considered immune escape as a function of the total days of (untreated) viral replication (Fig 2B). Interestingly, despite the >200 days of ART treatment, the accumulation of escape mutant virus followed a similar trajectory to that seen in previous primary infection studies [14, 27]. Furthermore, in both this and previous studies, decline in wild-type prevalence slowed after dropping to near 10% of the viral load and then appeared to persist in some animals at approximately 1% of the plasma virus. It is unclear why wild-type virus persists at these low levels despite the clear competitive advantage of escaped variants.

Despite the high levels of viral escape at the end of the first treatment interruption, proviral cell-associated DNA in LNMC and PBMC remained predominantly wild-type during the second round of ART (Fig 1D; days 303–323), suggesting only a small proportion of the reservoir was reseeded by the dominant virus circulating at the time of ART re-initiation. On average, LNMC and PBMC proviral DNA was 94% (n = 8) and 91% (n = 4) wild type during the second round of ART (ranges: 85–97% and 72–99%, respectively).

#### Reactivation of escape mutant virus early after second interruption

To investigate if prior treatment interruptions affect reactivation kinetics following subsequent ATIs, animals were recommenced on ART after the first ATI for a further 93 days, followed by a second interruption. Following the second ATI, high levels of immune escape were observed from the earliest sequencing timepoint (Figs 1D and 2A). Among animals sequenced within 6 days after rebound (n = 10) a median of 96% of virus was already escaped (range: 90% - 100%). This stands in stark contrast to the first interruption, where a median of 12% escape was observed in samples collected 10 days or less post rebound (p = 0.016).

The escape variants observed early during the second treatment interruption may have arisen from two possible sources. Firstly, wild-type virus may have reactivated from the latent reservoir, but then rapidly acquired de novo mutations and been selected due to immune pressure, leading to dominance of escape mutation in early samples. Alternatively, the escape variants generated during the first ATI may have directly reactivated from the reservoir and dominated early viral replication in the second ATI. To evaluate these hypotheses, we examined how well specific escape variants detected early in the second ATI were predicted by the corresponding recently replicating virus in the same animal (an 'escape reactivation' hypothesis). Alternatively, we considered whether WT virus may have reactivated and subsequently developed escape mutations de novo (a 'novel mutation' hypothesis). As illustrated in Fig 3, we found that variants detected early in the second ATI in each animal were best predicted by the variants replicating at the end of first ATI in the same animal (see S3 Text for details). Although the diversity of Tat-SL8 variants was broad during the first ATI, specific variants reached dominance at ART re-initiation in each animal. These dominant variants were then more likely to be detected in the rebounding viral load after the second ATI (indicated by red arrows). This strongly supports the hypothesis where escape variants in the second interruption arose from the escape variants that dominated in the first interruption.

#### Inhibition of wild-type Tat-SL8 reactivation from latency

The rapid appearance of escape variants during the second ATI could have occurred due to either higher reactivation rates of escaped virus compared to WT virus during the second ATI, or due to incomplete viral suppression during the shorter duration (3 months) of the second treatment phase. However, the available evidence strongly suggests a slower rate of successful reactivation in the second interruption than the first. Reactivation during the second



**Fig 3. Dynamics of the Tat-SL8 epitope variants.** The prevalence over time within animals RJy13 (**A**) and RUh14 (**B**) of wild-type and Tat-SL8 single amino acid variants resulting from a single non-synonymous nucleotide mutation. Each row corresponds to a different variant (wild type in the bottom row), and each column is a different sequencing time point. Percent of viral load composed of each variant indicated by colour bar. Red arrows indicate variants that were detected late in the first interruption and subsequently rebounded in the second treatment interruption.

https://doi.org/10.1371/journal.ppat.1011755.g003

interruption was delayed compared to the first interruption (Fig 1E; p = 0.019) and the frequency of reactivation was also around 6 times lower in the second interruption than in the first interruption (Fig 1G; median 0.55 per day vs 3.12, p = 0.0039). This argues against the possibility of a high reactivation rate of escaped virus or incomplete viral suppression. An alternative explanation, therefore, for the dominance of escape variants early in the second interruption, is immune 'sieving' of reactivating virus by Tat-SL8-specific T cells, resulting in suppression of wild-type virus rebound and thus leading to escape variants constituting the majority of reactivations observed.

To test this explanation, we performed a CD8 depletion in 10 animals following the second ATI (range: 90 to 97 days after the second ART interruption; approximately 465 days after initial infection) and assessed whether, once CD8<sup>+</sup> T cell pressure was removed, WT virus was permitted to rebound. For this depletion, the anti-CD8 $\alpha$  antibody MT807R1 was used, which also depletes Natural Killer Cells. Unsurprisingly, CD8 depletion had marked effects on plasma viral loads, including rapid increases in plasma viral levels in all animals (by mean of 1.79 log<sub>10</sub> copies/ml over first 4 days post depletion; Fig 4A), and detection of viral rebound in two animals that had previously demonstrated spontaneous control of viremia.

To analyse whether CD8 depletion affected the ability of WT virus to rebound, we compared viral lineages (i.e., barcodes) detected to have rebounded prior to CD8 depletion with those barcodes that were only seen after depletion (see Analysis definitions in Materials and Methods). We found strong evidence that CD8<sup>+</sup> T cell depletion radically altered the profile of rebounding virus. Firstly, there was a major difference in when the rebounding lineages were last observed in the replicating plasma virus pool (Fig 4B). Prior to CD8 depletion, the majority (64/104) of the rebounding barcodes were also present in plasma virus late in the first interruption (i.e., the most recent period of viral replication). In contrast, after CD8 depletion very few of the rebounding barcodes had been observed late in the first interruption (16/90), and indeed the majority of barcodes emerging after CD8 depletion (64/90) had been seen only in primary infection (day 7 or 14) or had not been sampled previously.



**Fig 4. Wild-type Tat-SL8 reactivates after CD8 depletion. (A)** Plasma viral loads pre- and post-CD8 depletion for 10 animals that were CD8 depleted during the second treatment interruption. (**B**) Breakdown of the origins of barcodes (i.e., last detection within plasma RNA) detected either before CD8 depletion or emerging after CD8 depletion. Fractional breakdown and total number for each prior time point indicated by pie charts and bar plot respectively (Pearson's chi-squared p-value provided). (**C**) Percent wild-type Tat-SL8 epitope at last detection prior to the second ATI. For barcodes never previously detected within the same animal (open circles), last detection was in the infecting inoculum, and therefore 100% wild-type Tat-SL8. P-value for association between percent wild-type at last detection and timing of detection in the second ATI is provided (Mann-Whitney test). (**D**) Reactivation rate estimates based on barcode ratios (see Estimation of the reactivation rate in Materials and Methods) for the first and second ATI, and post-CD8 depletion. For (**B-D**), barcodes seen in the second ATI that reactivated prior to and post-CD8 depletion are indicated by light red and purple, respectively.

https://doi.org/10.1371/journal.ppat.1011755.g004

The other major difference between the barcodes rebounding before or after CD8 depletion was whether those barcodes had previously been observed to be associated with escape mutations (in the plasma virus). Before CD8 depletion, most of rebounding barcodes had been observed to be escaped at last detection (65/104 barcodes were >50% escaped at the Tat-SL8 epitope). However, of barcodes emerging after CD8 depletion, only a minority were associated with majority escape variants at last detection (only 21/90 were >50% escaped, Fig 4C). In other words, sequencing of plasma virus after CD8<sup>+</sup> T cell depletion revealed the appearance of novel SIV barcodes that were mainly WT at the Tat-SL8 epitope when previously replicating. This strongly suggests that rebound of these lineages had been suppressed in the presence of CD8<sup>+</sup> T cell pressure, and only released after CD8 depletion.

The evidence above indicates that the detection of WT virus was greatly reduced among rebounding virus during the second ATI, likely as a result of Tat-SL8-specific CD8<sup>+</sup> T cell pressure blocking early reactivation or replication events of WT virus. This stands in contrast to the first treatment interruption, where early viral rebound was dominated by WT virus, suggesting WT virus was able to replicate to detectable levels despite the presence of Tat-SL8-specific CD8<sup>+</sup> T cells. It is important to differentiate between CD8<sup>+</sup> T cell effects on early viral reactivation events versus later viral replication and set point. For example, although WT virus dominated the rebound during the first ATI, the set point viral load was only a mean of 3.42  $\log_{10}$  copies/ml. This is substantially below the expected set point during primary infection of greater than 6 log<sub>10</sub> copies/ml [22, 28], suggesting substantial immune control. Okoye et al. have shown that rebound set points after 12 months of ART are around 2 logs lower than in primary infection, and that this is reversed by CD8 depletion. This suggests a major role of CD8s in controlling post-rebound set point viral levels [22]. This is also supported by the eventual dominance of escape mutant virus during the first ATI, suggesting selection pressure from Tat-SL8-specific CD8<sup>+</sup> T cells. Thus, although WT virus was able to successfully reactivate early in the first ATI, it also experienced substantial CD8<sup>+</sup> T cell pressure. This is consistent with Okoye et al.'s findings that CD8<sup>+</sup> T cell depletion during a first treatment interruption did not alter the early viral rebound kinetics but did alter the later set point viral levels [22]. In the second treatment interruption, however, Tat-SL8-specific CD8s controlled WT viral rebound to below the level of detection. The later set point level (of the predominantly EM viral population) was also lower than in the first treatment interruption (median of  $2.20 \log_{10}$ copies/ml), suggesting evidence of alternative control mechanisms. The strong suppression of WT virus during the second ATI suggests there was a quantitative or qualitative change in CD8<sup>+</sup> T cell function at the start of the second treatment interruption compared to the start of the first. Analysis of SL8-specific CD8<sup>+</sup> T cell concentration and phenotype (reported above) revealed minor differences, which do not seem sufficient to explain this apparent sieving of WT viral rebound in the second ATI. However, it seems likely that the response induced by 14 days of exposure to viral replication (before initial ART treatment) induced a different Tat-SL8-specific CD8<sup>+</sup> T cell response to that seen after an additional 60 days of treatment interruption.

The suppressive effect of CD8<sup>+</sup> T cells during the second treatment interruption is supported by not only the predominance of escape mutant virus, but also by the reduction in overall reactivation rate (from a median of 3.12/day to 0.55/day; p = 0.003), consistent with a >80% reduction in rebound frequency. The rebound frequency rose again after CD8<sup>+</sup> T cell depletion and was around 3-fold faster than the reactivation rate observed earlier in the second interruption (median 1.64 vs. 0.55/day; p = 0.059). Importantly, there was no significant difference between rebound rates after the first treatment interruption and after CD8<sup>+</sup> T cell depletion (median 3.12 vs. 1.64; p > 0.99), suggesting that in both cases this reflected SIV rebound kinetics in the absence of CD8<sup>+</sup> T cell sieving. Direct comparison of the number of rebounding barcodes following ATI 1, ATI 2, and after CD8 depletion reveals similar qualitative results as our estimation of reactivation rate (see S2 Fig)

## Discussion

It has long been known that CD8<sup>+</sup> T cells play an important role in the immune response during primary infection of PLWH and in SIV-infected RMs [29, 30]. However, the role of CD8<sup>+</sup> T cells during latency and viral reactivation from latency following treatment interruption are unclear. Two common hypotheses are that 1) CD8<sup>+</sup> T cells selectively restrict, or "sieve" the latent reservoir, blocking early events in the reactivation of sensitive (non-escaped) variants; or 2) CD8<sup>+</sup> T cell responses arise after initial virus reactivation has occurred, and the resultant selection pressure selects for de novo mutants during virus replication. In this study, we sought to infer the role of CD8<sup>+</sup> T cells during reactivation by monitoring the evolution of the Mamu-A\*01-restricted Tat-SL8 epitope over the course of successive treatment interruptions. Based on the relatively low frequency of successful reactivation events following the second treatment interruption, and the fact that reactivation was dominated by escape variants seen prior to treatment, we infer that CD8<sup>+</sup> T cells acted as a "sieve" of the latent reservoir to suppress early events in the reactivation of WT virus after the second treatment interruption. We note that the cell-associated viral DNA sampled prior to the second ATI was predominantly WT and that, despite this, escape variants dominated the observed reactivation. Given sequencing only identified clonotypes that had successfully grown to the level of detection by sequencing, other latently infected cells may have reactivated during the ATIs but either died out or failed to grow to the level of detection. It is not possible to differentiate whether CD8<sup>+</sup> T cell "sieving" of clonotypes occurs via early killing of the first reactivating cell, or control of subsequent viral replication such that a clonotype did not reach the level of detection.

The inferred "sieving" action of CD8<sup>+</sup> T cells is supported by the fact that subsequent CD8 depletion led to greater reactivation of the WT epitope, confirming the presence and competence of WT virus in the reactivating reservoir when not exposed to CD8 activity. It remains unclear why CD8<sup>+</sup> T cells appeared to permit the rebound of WT virus during the first ATI but appeared to suppress it so successfully during the second ATI. One possibility is that primary infection allowed only 14 days of viral replication before ART, and that this did not provide sufficient stimulation (and maturation) of responding CD8s. By contrast, the 60 days of viral replication during the first ATI may have boosted the quality and/or quantity of responding SL8-specific T cells. However, although the number of SL8-specific CD8<sup>+</sup> T cells in peripheral blood was mildly elevated prior to the second treatment interruption, this seems a surprisingly small change to have mediated suppression of WT rebound. Another possibility is that there was higher T cell activation at the time of the second interruption (for example because of the shorter (3 month) treatment). This again was not obvious from the T cell phenotype data, but comparison of phenotypes prior to ATIs was limited to peripheral blood due to sample availability. Hence, it is not clear if these data were fully representative of phenotypes present in the tissues. Further work is clearly required to delineate the numerical, phenotypic, and / or anatomical requirements that allow the SL8-specific T Cells to suppress early events in susceptible virus reactivation from latency.

Interestingly, the high frequency of immune escape from CD8 pressure during the second ATI did not lead to reduced virologic control compared to the first ATI. In fact, both peak and set point viral load were significantly lower at the second ATI than the first, potentially due to alternative means of immune pressure or the fitness costs of escape mutation [31]. Importantly, this suggests that immune escape during an ATI does not necessarily imply reduced immunologic control of viral replication during subsequent treatment interruptions.

It was previously suggested that CD8 pressure may drive viral evolution and escape during ART [17]. Our data strongly suggests that this is not the case given cell-associated proviral DNA remained predominantly wild-type during both rounds of ART and at both treatment

interruptions the viral quasispecies early after reactivation was dominated by virus previously observed in plasma.

A number of HIV studies have found that during ART, the composition of cell-associated DNA is biased towards virus that was circulating in the years prior to ART initiation [32-35]. We find that virus reactivating following treatment interruption is even more biased towards virus circulating at the time of ART (i.e., sampled within a matter of days prior to ART), as evidenced by the close agreement between barcodes and Tat-SL8 variants replicating at initiation of the second round of ART and those observed reactivating after the second treatment interruption. This was not because of a dominance of these variants in the cell-associated SIV DNA (observed during ART). Examination of the proviral DNA sequences before and after treatment interruption suggests that there was only slight reseeding of proviral DNA by replicating virus over the course of the first ATI. The vast majority of the proviral DNA observed during the second round of ART was established in primary infection (based on the high level of wildtype proviral DNA). Despite this, escape mutant virus observed during the first treatment interruption must have contributed to the rebound competent reservoir, given the strong relationship between virus circulating before and after the second round of ART (despite the fact that these mutants constitute only a minor fraction of the proviral DNA). The minimal reseeding of proviral DNA during treatment interruption is consistent with previous HIV/SIV studies [36-39]. Salantes et al. found significant differences in the composition of latent HIV-1 clones capable of producing virus on ART (i.e., results of the quantitative viral outgrowth assay on samples collected during ART) and those that were actually observed after treatment interruption (i.e., detected in replicating virus following ATI) [36], suggesting that even among replication competent proviral DNA, only a small subset is capable of initiating ongoing infection. Based on the detected rebounding of "older" virus following CD8 depletion in this study, we conjecture that susceptibility to immune pressure plays a substantial role in defining the component of proviral DNA that is most likely to initiate successful viral rebound.

One limitation of this study is that escape from  $CD8^+$  T cell recognition was not formally confirmed for all of the SL8 variants observed in this study. However, only 5 non-wild-type variants were ever detected at greater than 20% of the viral load in any animal (listed in S2 Table), and all of these mutants had previously been documented as escape variants and detected within Mamu-A\*01<sup>+</sup> RMs [14, 15, 27, 40]. We note that as stated above, CD8 depletion was performed using the anti-CD8 $\alpha$  antibody MT807R1, which also depletes Natural Killer Cells. However, the observations made here are highly epitope-specific, and it is not apparent that Natural Killer Cells can selectively target WT vs. escape mutant virus at the SL8 epitope. It should also be noted that in the comparison of viral lineages reactivating pre- vs. post-CD8 depletion, for some animals that reactivated pre-CD8 depletion, the most recent sequencing was performed over 60 days prior to CD8 depletion. Thus, some barcodes may have reactivated in the interim, causing an overestimation of the reactivation rate. However, 6 animals either had undetectable plasma virus for at least 30 days prior to CD8 depletion or had sequencing performed less than 20 days pre-CD8 depletion, giving greater confidence in the estimates presented here.

A final limitation of this study is it only examines the Tat-SL8 epitope, whereas there are many other CD8 T cell epitopes that have been documented in SIVmac239. Both antibodies and CD8<sup>+</sup> T cells specific to other epitopes likely play a role in reducing viral replication during ATI. However, it is not expected that these mechanisms, which target epitopes distinct from Tat-SL8, could disparately affect the different SL8 mutants that are the focus of our analysis. Thus, it is unlikely these other immune mechanisms were playing a role in the observed suppression of reactivation or early viral replication events of WT virus at the second interruption.

This study demonstrates that CD8 pressure can inhibit the early events in the reactivation of latent SIV/HIV, as evidenced by the slower reactivation rate and suppression of observed

WT reactivation at the second treatment interruption. In the SIV/Mamu-A\*01<sup>+</sup> NHP model used here, it seems likely that exposure to virus replication during the first ATI may have increased CD8<sup>+</sup> T cell function (while only modestly increasing SL8-specific cell numbers) leading to more effective control of WT virus during the second ATI. Notably, the exposure to viral replication during the first ATI may have acted as a double-edged sword, with replication providing antigen that boosted CD8 responses, but also facilitated viral escape from the same CD8 responses. These escape variants then seeded the reservoir and escaped CD8<sup>+</sup> T cell recognition at the subsequent ATI. However, this work suggests that strategies to boost SIV-specific CD8<sup>+</sup> T cell responses during ART (without stimulating viral escape) may permit the control of (at least wild-type) virus after treatment interruption.

## Materials and methods

#### **Ethics statement**

All procedures in this study were conducted following guidelines set forth by the Animal Welfare Act and by the NIH's Guide for the Care and Use of Laboratory Animals and performed in accordance with institutional regulations reviewed and approved by Emory University's Institutional Animal Care and Usage Committee (IACUC; Permit number: PROTO201700655) at Emory National Primate Research Center (YNPRC). Animal care facilities are accredited by the U.S. Department of Agriculture (USDA) and the Association for Assessment and Accreditation of Laboratory Animal Care (AAALAC) International. Appropriate procedures were performed to ensure that potential distress, pain, discomfort and/or injuries were limited to unavoidable level while conducting the research plan. Sedatives and analgesics were used when determined by veterinary medical staff for blood and tissue collections. Euthanasia of RMs was performed at the end of the study by veterinary medical staff, in accordance with IACUC endpoint guidelines.

#### SIV infection and antiretroviral therapy

This study was conducted on a cohort of 14 male, Mamu-A\*01<sup>+</sup> RMs between 46–129 months of age at infection. RMs were infected intravenously with 10,000 TCID50 of SIVmac239M, a genetically tagged virus with a 34-base genetic barcode inserted between the vpx and vpr accessory genes of the infectious molecular clone SIVmac239 (schematically illustrated in Fig 1B) [3]. All animals initiated a potent three drug ART regimen at 14 days post infection that included tenofovir (TDF; 5.1 mg/Kg per day), emtricitabine (FTC; 40 mg/Kg per day) and dolutegravir (DTG; 2.5 mg/Kg per day) formulated in a single daily injection (1ml/Kg per day; s.c.). ART was maintained until ~220 dpi, at which point the animals were stratified based on peak viral load, age, weight, and time to ART suppression of viremia and assigned to intervention/control groups. ART was interrupted for two months from approximately day 220 to day 282 post infection. FTY720 was orally administered to half the animals (n = 7) at a dose of 500µg/Kg per day beginning 1 month prior to ART interruption for a total of four months (from ~190 to ~310 dpi). Four animals were necropsied between 392 and 403 dpi. All animals underwent a second cycle of ART lasting three months (from ~282 to ~375 dpi). Ten animals underwent CD8<sup>+</sup> T cell depletion via a subcutaneous injection of 50mg/kg of MT807R1 antibody between 465 dpi and 471 dpi and were necropsied on day 571 or later.

## In vivo sample collection

Peripheral blood (PB) and lymph node (LN) biopsies were collected longitudinally. Blood collections were performed frequently during ATIs. Plasma was separated by density gradient centrifugation and used for quantification of viral RNA and flash frozen and used for viral sequencing. Peripheral blood mononuclear cells (PBMCs) were isolated from the remaining blood sample using a Ficoll-Paque Premium density gradient (GE Healthcare) and cryopreserved. The LN site over the axillary or inguinal region was clipped-off followed by an incision. The LN was then exposed upon dissection and excised over clamps. Each LN was segmented; one part was flash frozen in dry ice for RNA-Seq or for studying the pro-viral compartment (Cell-associated DNA/RNA) of SIV, and the other part was mechanically disrupted through a 70 µm cell strainer and washed with R-10 media composed of RPMI 1640 (Corning) supplemented with 10% heat-inactivated fetal bovine serum (FBS), 100 IU/ml penicillin, 100 µg/ml streptomycin, and 200 mM L-glutamine (GeminiBio) to obtain mononuclear cells that were cryopreserved for flow cytometry analysis.

#### Plasma viral load quantification

Plasma SIVmac239M loads were quantified regularly throughout the study at the Virology Core of the Emory Center for AIDS Research using a standard quantitative PCR (qPCR) assay (with a limit of detection of 60 copies/ml) as described previously [41].

#### **Barcode and Tat-SL8 sequencing**

RNA was extracted from plasma samples using QIAGEN EZ1 Virus Mini Kit v2.0 and cDNA was synthesized using Superscript III reverse transcriptase and a reverse primer (SL8R1: 5'-AGC TGA GAG AGG ATT TCC TCCC-3'). DNA was extracted from mononuclear cells using QIAamp DNA Blood Mini Kit. qRT-PCR was used to quantify the number of templates using SYBR Green and primers (VpxF1: 5'-CTA GGG GAA GGA CAT GGG GCA GG-3' and VprR1: 5'-CCA GAA CCT CCA CTA CCC ATT CATC-3'). DNA was amplified and MiSeq adaptors were added directly to the amplicons using forward primer VpxF1.Illumina: 5'-TCG TCG GCA GCG TCA GAT GTG TAT AAG AGA CAG CTA GGG GAA GGA CAT GGG GCA GG-3' and reverse primer SL8R1.Illumina: 5'-GTC TCG TGG GCT CGG AGA TGT GTA TAA GAG ACA GAG CTG AGA GAG GAT TTC CTC CC-3'. PCR conditions included an initial denaturation of 2 minutes at 94°C, followed by 40 cycles of denaturation (94°C, 15 sec), annealing (60°C, 1 min 30 sec), and extension (68°C, 1 min), followed by a final extension step of 5 minutes at 68°C. Reactions were pooled and gel extracted using QIAquick Gel Extraction Kit to remove primer dimers and sequenced with Illumina Next Generation Sequencing approach.

Sequencing errors are relatively common in Illumina Next Generation Sequencing, and therefore to be confident in the results, output reads were discarded if the barcode did not match a barcode from the previous stock characterization, the Tat-SL8 epitope was not detected, or an indel was detected in the Tat-SL8 epitope. Number of templates and sequence amplification varied between sampling time points, and sequencing runs with less than 1000 templates (200) or 200 reads (100) for RNA (DNA) after the removal of erroneous reads were discarded from further analysis. Limit of detection was set for each sequencing run based on number of templates and output reads (see <u>S1 Text</u> for details). Due to differences in limit of detection for various barcodes and Tat-SL8 variants at different time points, for quantification measurements, background subtraction was performed on all barcodes, Tat-SL8 variants, and barcode-Tat-SL8 combinations detected unless stated otherwise.

### T cell quantification and phenotype characterization

**MHC Class I Tat-TL8 Tetramer.** The Mamu-A\*01 Tat28-35 APC conjugated tetramer was constructed using an  $SIV_{MAC251}$ -derived peptide (TTPESANL) and was supplied by the

NIH Tetramer core facility at Emory University, Atlanta, GA. Even though the corresponding  $SIV_{MAC239}$  sequence was STPESANL, this tetramer detected strong responses in  $SIV_{MAC239}$ -infected macaques, as it has been shown before where  $Tat_{28-35}$  STPESANL tetramer staining yielded identical results [14].

Fluorescence cytometry staining. The following antibodies were obtained from BD Biosciences: CD4 BUV661 (clone SK3), CD95 BUV737 (clone DX2) and CD3 BUV805 (clone SP34-2); from Biolegend: CD8 BV650 (clone RPA-T8), CD14 BV510 (clone M5E2), CD16 BV510 (clone 3G8) and CD20 BV510 (clone 2H7); from Beckman Coulter: CD28 ECD (clone CD28.2). Briefly, cryopreserved PBMC were thawed and stained with MHC-Class I tetramer for 10 minutes. Then, cells were stained for viability exclusion using Live/Dead Fixable Aqua (Invitrogen) for 10 minutes, followed by a 20-minute incubation with a panel of directly conjugated monoclonal antibodies diluted in equal parts of fluorescence-activated cell sorting (FACS) buffer (PBS containing 0.1% sodium azide and 1% bovine serum albumin) and Brilliant stain buffer (BD Biosciences, San Jose, CA). Cells were washed in FACS buffer and then fixed/permeabilized using the FoxP3 Transcription Factor Buffer Kit (eBioscience, San Diego, CA). Intracellular staining was performed by adding the antibody cocktail prepared in 1X perm-wash buffer for 1 hour. Incubations were all done at room temperature. Stained cells were washed and fixed in PBS containing 1% paraformaldehyde (Sigma- Aldrich, St. Luis, MO). Samples were acquired using a FACS Symphony A5 cytometer and analysis was performed by Flowjo software 10.8.1 (Tree Star Inc.).

#### Statistical and mathematical analysis

Unless otherwise stated in the text or following Methods, all statistical tests for comparisons of measurements at two time points were performed using the paired Wilcoxon signed rank test.

Estimation of the reactivation rate. In order to estimate the reactivation rate in an animal (*R*), we use the same method as in our previous work [2, 3]. Briefly, we estimate the reactivation rate using the average log-ratio of barcodes, and assuming that all barcodes have the same growth rate (*g*) during initial exponential growth of the virus and that the growth rate of a single barcode is the same as the growth rate of total viral load. We estimate the delay between reactivations of two barcodes using the growth rate and the log-ratio of the barcodes  $(\ln S_2 - \ln S_1)$  as  $\Delta = \frac{\ln S_2 - \ln S_1}{g}$ . Therefore, the estimated reactivation rate based on *n* rebounding barcodes is the inverse of the average of *n*-1 delays between reactivations:

$$R = \frac{g(n-1)}{\sum_{i=1}^{n-1} (\ln S_{i+1} - \ln S_i)},$$
(1)

where  $S_i$  is the number of reads for barcode *i*, for i = 1, 2, ..., n. To estimate the viral growth rate in an animal (*g*), we first calculated the growth rate of the virus between each pair of neighboring measurements (timepoints  $t_1$  and  $t_{l+1}$ ) following rebound of virus in said animal. This included the interval from the last timepoint at which the viral load was below the detection threshold in the animal (for which viral load is approximated as detection threshold value) and the first measurement at which the viral load is above the detection threshold in. Therefore, we have:

$$g_l = \frac{\ln V(t_{l+1}) - \ln V(t_l)}{t_{l+1} - t_l}, \qquad l = 1, 2, \dots$$

where  $V(t_l)$  is the viral load at the time  $t_l$ . The growth rate g for the animal is assumed to be the maximal growth rate between any two neighboring measurements:

$$g = \max_{l} g_{l}$$

The procedure of estimation of reactivation rate was implemented in Wolfram Mathematica, Wolfram Research Inc, Champaign, IL, USA. The paired Wilcoxon signed rank test (GraphPad Prism 9.4.1) was used to compare reactivation rate following the first and second ATI.

For estimation of reactivation rate following CD8 depletion, the same methodology was implemented, except only including barcodes deemed to have reactivated after CD8 depletion (see Analysis definitions below). Comparison of reactivation rates across the three time points; first ATI, second ATI, and CD8 depletion, was done using the Kruskal-Wallis test with Dunn's multiple comparisons (GraphPad Prism 9.4.1).

**Estimation of time-to-detection.** We used GraphPad Prism 9.4.1 to obtain the Kaplan-Meier curves for the time to detected reactivation and compared the time to detected reactivation following the two ATIs using the log-rank test.

The time to detected reactivation was considered to be the earliest of 1) the first of at least two viral load quantifications above limit of detection or 2) viral load quantification of above 1000 copies/ml on a single day. Day post ATI was calculated by defining the first day without ART as day 0 post ATI. In the case that ART was resumed, CD8s were depleted, or animal was necropsied before reactivation was detected, animals were right censored at the day of said event.

Predictors of plasma virus variant composition post treatment interruption. To assess the factors contributing to the detection of a variant in plasma virus after treatment interruption, we consider different potential mechanisms for generating escape mutations in the rebounding virus; One possibility is that the initial rebounding virus is actually wild type, but that mutation and selection during early replication led to the dominance of EM virus by the time we sample plasma. In this case, the mutants we see might be a random selection of possible mutants (e.g., all mutants previously observed in any animal have equal possibility of being seen). Or, more likely, there is a hierarchy of likely escape mutations (determined by the mutation probability, fitness etc.), but these probabilities are similar across different animals (so the probability of a given mutation is similar across animals). An alternative scenario is that escape mutant virus (previously observed in plasma in a given animal) rebounded from the reservoir. In this case, the rebounding virus in a given animal should resemble both the mutations and barcodes observed earlier in infection. To test this, we fit a Poisson model of different sources of viral variability early following the 2<sup>nd</sup> ATI; 1) equal probability of all mutants, 2) observed hierarchy of mutants across all animals, or 3) hierarchy of mutations seen at the time of the second ART in the same animal, to see which best predicted the escape mutants present in plasma. We then explored whether a combination of two mechanisms improved the fit (hypothesis 4). As derived in S3 Text, the probability of variant k being detected following rebound in animal  $j(z_{k,j})$  can be modeled as

$$z_{kj}(\overrightarrow{x}_{kj}; \overrightarrow{\theta}_j) = 1 - e^{-r_{kj}(\overrightarrow{x}_{kj}; \overrightarrow{\theta}_j)}$$
(2)

where  $r_{k,j}$  is a function of both variant specific and animal specific variables,  $\vec{x}_{k,j}$ , and animal specific parameters,  $\vec{\theta}_j$ , and takes on a different form depending on the hypothesis being considered (see <u>S3 Text</u> for the 4 forms of  $r_{k,j}(\vec{x}_{k,j}; \vec{\theta}_j)$ ).

For each hypothesis (and therefore each form of  $r_{k,j}$ ), Eq (2) was optimized to the data from each animal using a maximum likelihood approach with the log-likelihood for animal *j* defined by:

$$\ln \mathcal{L}_{j}(\overrightarrow{\theta}_{j}) = \sum_{k} d_{kj} \ln(z_{kj}(\overrightarrow{x}_{kj}; \overrightarrow{\theta}_{j})) + (1 - d_{kj}) \ln(1 - z_{kj}(\overrightarrow{x}_{kj}; \overrightarrow{\theta}_{j}))$$
(3)

where  $d_{k,j}$  is an indicator variable for if variant k was detected (1) or not (0) in the reactivation pool of animal j, and  $\overrightarrow{\theta}_j$  and  $\overrightarrow{x}_{k,j}$  are as defined above. The Akaike Information Criterion [42] and likelihood ratio tests were used to compare the relative support for hypotheses 1 to 4 listed above and hence, whether it is most likely that the variants detected following the second treatment interruption were the result of reactivation from the reservoir and/or mutation from reactivated wild-type virus.

**Analysis definitions.** Peak viral load is defined as the maximum viral load within 30 days of detected viral rebound.

The mean time weighted  $\log_{10}$  viral set point is calculated based on viral loads between 30 days post rebound detection and 60 days post rebound detection, resumption of ART, CD8 depletion, or necropsy, whichever came first. That is, mean time weighted  $\log_{10}$  viral set point is

$$\frac{1}{t_f - t_i} \int_{t_i}^{t_f} \log_{10} V(t) dt$$

where V(t), is the viral load at time t,  $t_i$  is the time of the first viral load quantification 30 days or more after rebound detection, and  $t_f$  is the minimum of 1) time of last viral load quantification within 60 days of rebound detection, 2) time of ART resumption, 3) time of CD8 depletion, and 4) necropsy. For a number of animals,  $t_i = t_f$  and in these instances, mean time weighted  $\log_{10}$  viral set point was set as  $\log_{10} V(t_i)$ . It is assumed that  $\log_{10} V(t)$  changed linearly between viral load measurements.

Fold increase in viral load following CD8 depletion, is the ratio of first viral load quantification after the day of CD8 depletion to that on the day of CD8 depletion.

When required in calculations, below detection viral loads were set to the limit of detection (60 copies/ml).

Barcodes were deemed to have reactivated post-CD8 depletion as follows. 1) In the 6 animals with sequencing done both prior to and after CD8 depletion, barcodes detected post-CD8 depletion but not earlier in the second interruption were categorized as reactivating post-CD8 depletion. 2) In 3 animals sequenced post-CD8 depletion but not prior (2 animals that did not reactivate prior to CD8 depletion and one that had undetectable viral load for 30 days prior to CD8 depletion), all barcodes detected post-CD8 depletion were categorized as reactivating post-CD8 depletion.

**Statistical packages.** Statistical tests were performed in MATLAB (MathWorks, R2021b), GraphPad Prism 9.4.1, or R version 4.0.3 run through RStudio.

## Supporting information

**S1 Text. Limit of detection for Illumina sequencing.** (DOCX)

**S2 Text. Tat-SL8-specific CD8<sup>+</sup> T cell phenotype flow cytometry analysis.** (DOCX)

**S3 Text.** Modeling predictors of variant reactivation at second treatment interruption. (DOCX)

S1 Table. Tat-SL8-specific CD8<sup>+</sup> T Cell Phenotype Comparison: Prior to 1<sup>st</sup> ATI vs. Prior to 2<sup>nd</sup> ATI (day ~190 vs. 373–374). (DOCX) S2 Table. Non-wild-type Tat-SL8 variants detected at greater than 20% of viral load at least once.

(DOCX)

**S1** Fig. Gating strategy used for flow cytometric analyses. (DOCX)

**S2 Fig. Rebounding barcodes.** (DOCX)

## **Author Contributions**

- **Conceptualization:** Steffen S. Docken, Brandon F. Keele, Mirko Paiardini, Michael R. Betts, Katharine J. Bar, Miles P. Davenport.
- Data curation: Steffen S. Docken, Kevin McCormick.
- Formal analysis: Steffen S. Docken, Mykola Pinkevych.
- **Investigation:** Kevin McCormick, M. Betina Pampena, Sadia Samer, Emily Lindemuth, Elise G. Viox.
- **Methodology:** Steffen S. Docken, Mykola Pinkevych, Elise G. Viox, Yuhuang Wu, Timothy E. Schlub, Deborah Cromer, Mirko Paiardini, Michael R. Betts, Katharine J. Bar, Miles P. Davenport.
- Resources: Mirko Paiardini, Michael R. Betts, Katharine J. Bar, Miles P. Davenport.
- Writing original draft: Steffen S. Docken, M. Betina Pampena, Elise G. Viox, Miles P. Davenport.
- Writing review & editing: Steffen S. Docken, M. Betina Pampena, Elise G. Viox, Deborah Cromer, Brandon F. Keele, Mirko Paiardini, Michael R. Betts, Katharine J. Bar, Miles P. Davenport.

#### References

- Pinkevych M, Cromer D, Tolstrup M, Grimm AJ, Cooper DA, Lewin SR, et al. HIV Reactivation from Latency after Treatment Interruption Occurs on Average Every 5–8 Days—Implications for HIV Remission. PLoS Pathog. 2015; 11(7):e1005000. Epub 2015/07/03. https://doi.org/10.1371/journal.ppat. 1005000 PMID: 26133551; PubMed Central PMCID: PMC4489624.
- Pinkevych M, Fennessey CM, Cromer D, Reid C, Trubey CM, Lifson JD, et al. Predictors of SIV recrudescence following antiretroviral treatment interruption. Elife. 2019; 8. Epub 2019/10/28. https://doi.org/ 10.7554/eLife.49022 PMID: 31650954; PubMed Central PMCID: PMC6917497.
- Fennessey CM, Pinkevych M, Immonen TT, Reynaldi A, Venturi V, Nadella P, et al. Genetically-barcoded SIV facilitates enumeration of rebound variants and estimation of reactivation rates in nonhuman primates following interruption of suppressive antiretroviral therapy. PLoS Pathog. 2017; 13(5): e1006359. Epub 2017/05/05. https://doi.org/10.1371/journal.ppat.1006359 PMID: 28472156; PubMed Central PMCID: PMC5433785.
- Scheid JF, Horwitz JA, Bar-On Y, Kreider EF, Lu CL, Lorenzi JC, et al. HIV-1 antibody 3BNC117 suppresses viral rebound in humans during treatment interruption. Nature. 2016; 535(7613):556–60. Epub 20160622. https://doi.org/10.1038/nature18929 PMID: 27338952; PubMed Central PMCID: PMC5034582.
- Bar KJ, Sneller MC, Harrison LJ, Justement JS, Overton ET, Petrone ME, et al. Effect of HIV Antibody VRC01 on Viral Rebound after Treatment Interruption. N Engl J Med. 2016; 375(21):2037–50. Epub 20161109. https://doi.org/10.1056/NEJMoa1608243 PMID: 27959728; PubMed Central PMCID: PMC5292134.
- Cohen YZ, Lorenzi JCC, Krassnig L, Barton JP, Burke L, Pai J, et al. Relationship between latent and rebound viruses in a clinical trial of anti-HIV-1 antibody 3BNC117. J Exp Med. 2018; 215(9):2311–24.

Epub 2018/08/04. https://doi.org/10.1084/jem.20180936 PMID: 30072495; PubMed Central PMCID: PMC6122972.

- Julg B, Stephenson KE, Wagh K, Tan SC, Zash R, Walsh S, et al. Safety and antiviral activity of triple combination broadly neutralizing monoclonal antibody therapy against HIV-1: a phase 1 clinical trial. Nat Med. 2022; 28(6):1288–96. Epub 20220512. https://doi.org/10.1038/s41591-022-01815-1 PMID: 35551291; PubMed Central PMCID: PMC9205771.
- Burwitz BJ, Giraldo-Vela JP, Reed J, Newman LP, Bean AT, Nimityongskul FA, et al. CD8+ and CD4+ cytotoxic T cell escape mutations precede breakthrough SIVmac239 viremia in an elite controller. Retrovirology. 2012; 9:91. Epub 2012/11/08. <u>https://doi.org/10.1186/1742-4690-9-91</u> PMID: 23131037; PubMed Central PMCID: PMC3496649.
- McBrien JB, Kumar NA, Silvestri G. Mechanisms of CD8(+) T cell-mediated suppression of HIV/SIV replication. Eur J Immunol. 2018; 48(6):898–914. Epub 2018/02/11. <u>https://doi.org/10.1002/eji.</u> 201747172 PMID: 29427516; PubMed Central PMCID: PMC6531861.
- O'Connor DH, McDermott AB, Krebs KC, Dodds EJ, Miller JE, Gonzalez EJ, et al. A dominant role for CD8+-T-lymphocyte selection in simian immunodeficiency virus sequence variation. J Virol. 2004; 78 (24):14012–22. Epub 2004/11/27. https://doi.org/10.1128/JVI.78.24.14012-14022.2004 PMID: 15564508; PubMed Central PMCID: PMC533930.
- Leslie AJ, Pfafferott KJ, Chetty P, Draenert R, Addo MM, Feeney M, et al. HIV evolution: CTL escape mutation and reversion after transmission. Nat Med. 2004; 10(3):282–9. Epub 2004/02/11. <u>https://doi.org/10.1038/nm992</u> PMID: 14770175.
- Goulder PJ, Phillips RE, Colbert RA, McAdam S, Ogg G, Nowak MA, et al. Late escape from an immunodominant cytotoxic T-lymphocyte response associated with progression to AIDS. Nat Med. 1997; 3 (2):212–7. Epub 1997/02/01. https://doi.org/10.1038/nm0297-212 PMID: 9018241.
- Feeney ME, Tang Y, Roosevelt KA, Leslie AJ, McIntosh K, Karthas N, et al. Immune escape precedes breakthrough human immunodeficiency virus type 1 viremia and broadening of the cytotoxic T-lymphocyte response in an HLA-B27-positive long-term-nonprogressing child. J Virol. 2004; 78(16):8927–30. Epub 2004/07/29. https://doi.org/10.1128/JVI.78.16.8927-8930.2004 PMID: 15280502; PubMed Central PMCID: PMC479057.
- Allen TM, O'Connor DH, Jing P, Dzuris JL, Mothe BR, Vogel TU, et al. Tat-specific cytotoxic T lymphocytes select for SIV escape variants during resolution of primary viraemia. Nature. 2000; 407 (6802):386–90. Epub 2000/10/03. https://doi.org/10.1038/35030124 PMID: 11014195.
- Price DA, West SM, Betts MR, Ruff LE, Brenchley JM, Ambrozak DR, et al. T cell receptor recognition motifs govern immune escape patterns in acute SIV infection. Immunity. 2004; 21(6):793–803. <u>https:// doi.org/10.1016/j.immuni.2004.10.010</u> PMID: 15589168.
- Loffredo JT, Burwitz BJ, Rakasz EG, Spencer SP, Stephany JJ, Vela JP, et al. The antiviral efficacy of simian immunodeficiency virus-specific CD8+ T cells is unrelated to epitope specificity and is abrogated by viral escape. J Virol. 2007; 81(6):2624–34. Epub 2006/12/29. <u>https://doi.org/10.1128/JVI.01912-06</u> PMID: 17192314; PubMed Central PMCID: PMC1866004.
- Casazza JP, Betts MR, Hill BJ, Brenchley JM, Price DA, Douek DC, et al. Immunologic pressure within class I-restricted cognate human immunodeficiency virus epitopes during highly active antiretroviral therapy. J Virol. 2005; 79(6):3653–63. https://doi.org/10.1128/JVI.79.6.3653-3663.2005 PMID: 15731259; PubMed Central PMCID: PMC1075692.
- Karlsson AC, Chapman JM, Heiken BD, Hoh R, Kallas EG, Martin JN, et al. Antiretroviral drug therapy alters the profile of human immunodeficiency virus type 1-specific T-cell responses and shifts the immunodominant cytotoxic T-lymphocyte response from Gag to Pol. J Virol. 2007; 81(20):11543–8. Epub 2007/08/03. https://doi.org/10.1128/JVI.00779-07 PMID: <u>17670829</u>; PubMed Central PMCID: PMC2045537.
- Oxenius A, Price DA, Gunthard HF, Dawson SJ, Fagard C, Perrin L, et al. Stimulation of HIV-specific cellular immunity by structured treatment interruption fails to enhance viral control in chronic HIV infection. Proc Natl Acad Sci U S A. 2002; 99(21):13747–52. Epub 2002/10/09. https://doi.org/10.1073/ pnas.202372199 PMID: 12370434; PubMed Central PMCID: PMC129766.
- Papasavvas E, Ortiz GM, Gross R, Sun J, Moore EC, Heymann JJ, et al. Enhancement of human immunodeficiency virus type 1-specific CD4 and CD8 T cell responses in chronically infected persons after temporary treatment interruption. J Infect Dis. 2000; 182(3):766–75. Epub 2000/08/19. <u>https://doi.org/10.1086/315748</u> PMID: 10950770.
- Rosenberg ES, Altfeld M, Poon SH, Phillips MN, Wilkes BM, Eldridge RL, et al. Immune control of HIV-1 after early treatment of acute infection. Nature. 2000; 407(6803):523–6. Epub 2000/10/12. <a href="https://doi.org/10.1038/35035103">https://doi.org/10.1038/35035103</a> PMID: 11029005.
- Okoye AA, Duell DD, Fukazawa Y, Varco-Merth B, Marenco A, Behrens H, et al. CD8+ T cells fail to limit SIV reactivation following ART withdrawal until after viral amplification. J Clin Invest. 2021; 131(8).

Epub 2021/02/26. https://doi.org/10.1172/JCI141677 PMID: 33630764; PubMed Central PMCID: PMC8262469.

- Pinkevych M, Docken SS, Okoye AA, Fennessey CM, Del Prete GQ, Pino M, et al. Timing of initiation of anti-retroviral therapy predicts post-treatment control of SIV replication. PLoS Pathog. 2023; 19(10): e1011660. Epub 2023/10/06. https://doi.org/10.1371/journal.ppat.1011660 PMID: 37801446; PubMed Central PMCID: PMC10558076.
- Mandala S, Hajdu R, Bergstrom J, Quackenbush E, Xie J, Milligan J, et al. Alteration of lymphocyte trafficking by sphingosine-1-phosphate receptor agonists. Science. 2002; 296(5566):346–9. Epub 2002/ 03/30. https://doi.org/10.1126/science.1070238 PMID: 11923495.
- Brinkmann V, Davis MD, Heise CE, Albert R, Cottens S, Hof R, et al. The immune modulator FTY720 targets sphingosine 1-phosphate receptors. J Biol Chem. 2002; 277(24):21453–7. Epub 2002/04/23. https://doi.org/10.1074/jbc.C200176200 PMID: 11967257.
- Pino M, Pagliuzza A, Pampena MB, Deleage C, Viox EG, Nguyen K, et al. Limited impact of fingolimod treatment during the initial weeks of ART in SIV-infected rhesus macaques. Nat Commun. 2022; 13 (1):5055. Epub 2022/08/28. https://doi.org/10.1038/s41467-022-32698-y PMID: 36030289; PubMed Central PMCID: PMC9420154.
- Immonen TT, Camus C, Reid C, Fennessey CM, Del Prete GQ, Davenport MP, et al. Genetically barcoded SIV reveals the emergence of escape mutations in multiple viral lineages during immune escape. Proc Natl Acad Sci U S A. 2020; 117(1):494–502. Epub 2019/12/18. https://doi.org/10.1073/pnas. 1914967117 PMID: 31843933; PubMed Central PMCID: PMC6955354.
- Allen TM, Mortara L, Mothe BR, Liebl M, Jing P, Calore B, et al. Tat-vaccinated macaques do not control simian immunodeficiency virus SIVmac239 replication. J Virol. 2002; 76(8):4108–12. Epub 2002/03/22. https://doi.org/10.1128/jvi.76.8.4108-4112.2002 PMID: <u>11907251</u>; PubMed Central PMCID: PMC136098.
- 29. Klatt NR, Shudo E, Ortiz AM, Engram JC, Paiardini M, Lawson B, et al. CD8+ lymphocytes control viral replication in SIVmac239-infected rhesus macaques without decreasing the lifespan of productively infected cells. PLoS Pathog. 2010; 6(1):e1000747. Epub 20100129. https://doi.org/10.1371/journal.ppat.1000747 PMID: 20126441; PubMed Central PMCID: PMC2813271.
- Wong JK, Strain MC, Porrata R, Reay E, Sankaran-Walters S, Ignacio CC, et al. In vivo CD8+ T-cell suppression of siv viremia is not mediated by CTL clearance of productively infected cells. PLoS Pathog. 2010; 6(1):e1000748. Epub 20100129. https://doi.org/10.1371/journal.ppat.1000748 PMID: 20126442; PubMed Central PMCID: PMC2813272.
- Loh L, Batten CJ, Petravic J, Davenport MP, Kent SJ. In vivo fitness costs of different Gag CD8 T-cell escape mutant simian-human immunodeficiency viruses for macaques. J Virol. 2007; 81(10):5418–22. Epub 20070307. https://doi.org/10.1128/JVI.02763-06 PMID: 17344299; PubMed Central PMCID: PMC1900239.
- Brodin J, Zanini F, Thebo L, Lanz C, Bratt G, Neher RA, et al. Establishment and stability of the latent HIV-1 DNA reservoir. Elife. 2016;5. Epub 2016/11/18. <u>https://doi.org/10.7554/eLife.18889</u> PMID: 27855060; PubMed Central PMCID: PMC5201419.
- Abrahams MR, Joseph SB, Garrett N, Tyers L, Moeser M, Archin N, et al. The replication-competent HIV-1 latent reservoir is primarily established near the time of therapy initiation. Sci Transl Med. 2019; 11(513). https://doi.org/10.1126/scitranslmed.aaw5589 PMID: <u>31597754</u>; PubMed Central PMCID: PMC7233356.
- Omondi FH, Sudderuddin H, Shahid A, Kinloch NN, Jones BR, Miller RL, et al. HIV Proviral Burden, Genetic Diversity, and Dynamics in Viremic Controllers Who Subsequently Initiated Suppressive Antiretroviral Therapy. mBio. 2021; 12(6):e0249021. Epub 2021/11/17. https://doi.org/10.1128/mBio.02490-21 PMID: 34781741; PubMed Central PMCID: PMC8693448.
- Gondim MVP, Sherrill-Mix S, Bibollet-Ruche F, Russell RM, Trimboli S, Smith AG, et al. Heightened resistance to host type 1 interferons characterizes HIV-1 at transmission and after antiretroviral therapy interruption. Sci Transl Med. 2021; 13(576). https://doi.org/10.1126/scitranslmed.abd8179 PMID: 33441429; PubMed Central PMCID: PMC7923595.
- Salantes DB, Zheng Y, Mampe F, Srivastava T, Beg S, Lai J, et al. HIV-1 latent reservoir size and diversity are stable following brief treatment interruption. J Clin Invest. 2018; 128(7):3102–15. Epub 2018/06/ 19. https://doi.org/10.1172/JCI120194 PMID: 29911997; PubMed Central PMCID: PMC6026010.
- Immonen TT, Fennessey CM, Lipkey L, Thorpe A, Del Prete GQ, Lifson JD, et al. Transient viral replication during analytical treatment interruptions in SIV infected macaques can alter the rebound-competent viral reservoir. PLoS Pathog. 2021; 17(6):e1009686. Epub 2021/06/19. https://doi.org/10.1371/journal. ppat.1009686 PMID: 34143853; PubMed Central PMCID: PMC8244872.
- **38.** Clarridge KE, Blazkova J, Einkauf K, Petrone M, Refsland EW, Justement JS, et al. Effect of analytical treatment interruption and reinitiation of antiretroviral therapy on HIV reservoirs and immunologic

parameters in infected individuals. PLoS Pathog. 2018; 14(1):e1006792. Epub 2018/01/13. https://doi.org/10.1371/journal.ppat.1006792 PMID: 29324842; PubMed Central PMCID: PMC5764487.

- Strongin Z, Sharaf R, VanBelzen DJ, Jacobson JM, Connick E, Volberding P, et al. Effect of Short-Term Antiretroviral Therapy Interruption on Levels of Integrated HIV DNA. J Virol. 2018; 92(12). Epub 2018/ 03/30. https://doi.org/10.1128/JVI.00285-18 PMID: 29593048; PubMed Central PMCID: PMC5974505.
- 40. Moriarty RV, Golfinos AE, Gellerup DD, Schweigert H, Mathiaparanam J, Balgeman AJ, et al. The mucosal barrier and anti-viral immune responses can eliminate portions of the viral population during transmission and early viral growth. PLoS One. 2021; 16(12):e0260010. Epub 20211202. https://doi.org/10.1371/journal.pone.0260010 PMID: 34855793; PubMed Central PMCID: PMC8639003.
- 41. Taaffe J, Chahroudi A, Engram J, Sumpter B, Meeker T, Ratcliffe S, et al. A five-year longitudinal analysis of sooty mangabeys naturally infected with simian immunodeficiency virus reveals a slow but progressive decline in CD4+ T-cell count whose magnitude is not predicted by viral load or immune activation. J Virol. 2010; 84(11):5476–84. Epub 20100324. https://doi.org/10.1128/JVI.00039-10 PMID: 20335252; PubMed Central PMCID: PMC2876601.
- **42.** Burnham KP, Anderson DR. Information and Likelihood Theory: A Basis for Model Selection and Inference. Model selection and multimodel inference: a practical information-theoretic approach. 2 ed. New York: Springer; 2002. p. 49–97.